\section{bib}

\cite{TrackEffRun1}  
\cite{PDG}  
\cite{ROOT}
\cite{sPLOT1}
\cite{sPLOT2}


\section{SM}

\cite{SM-Glashow-1}
\cite{SM-Weinberg}
\cite{SM-Salam}
\cite{SM-Salam-2}
\cite{SM-Peskin}

\cite{SM-GIM}
\cite{SM-Sakharov}
\cite{SM-Wolfenstein}
\cite{SM-CKM}
\cite{SM-Cabibbo}

\cite{SM-Quigg}
\cite{SM-Noether}   
\cite{SM-Higgs}
\cite{SM-Englert}
\cite{SM-Conserve}

\cite{SM-KM}
\cite{SM-leptoquarks}
\cite{SM-Thompson}
\cite{SM-Rutherford}
\cite{SM-Chadwick}

\cite{SM-Anderson}
\cite{SM-Becquerel}
\cite{SM-Fermi}
\cite{SM-Fermi-reject}
\cite{SM-Gamow}

\cite{SM-Lee}
\cite{SM-Wu}
\cite{SM-Lederman}
\cite{SM-Friedman}
\cite{SM-Dalitz}

\cite{SM-Feynman}
\cite{SM-Sudarshan}
\cite{SM-Weyl}  
\cite{SM-Lee-2}
\cite{SM-Okun}  

\cite{SM-Appelquist}    
\cite{SM-GellMann-1}
\cite{SM-Yang-Mills}
\cite{SM-Goldstone}
\cite{SM-Goldstone-2}

\cite{SM-Guralnik}
\cite{SM-Gell-Mann-2}
\cite{SM-Neeman}
\cite{SM-Gell-Mann-3}
\cite{SM-Zweig-1}

\cite{SM-Zweig-2}
\cite{SM-Bjorken}
\cite{SM-Cronin}
\cite{SM-Gaillard}
\cite{SM-Buchalla}

\cite{SM-LHCb_BR}
\cite{SM-QCDF}
\cite{SM-OPE}
\cite{NP-portals}
\cite{NP-portals2}

\cite{NP-portals3}
\cite{NP-loops}
\cite{NP-loops2}
\cite{NP-loops3}
\cite{NP-Leptoquarks0}

\cite{NP-Leptoquarks}
\cite{NP-Leptoquarks2}
\cite{NP-Leptoquarks3}
\cite{NP-Leptoquarks4}
\cite{SM-Wilson}

\cite{SM-LCSM}
\cite{SM-CharmLoop}
\cite{SM-leptoquarkportal-1}
\cite{SM-leptoquarkportal-2}
\cite{SM-leptoquarkportal-3}

\section{Experiment}

\cite{LHChistory}
\cite{LEPstory}
\cite{LHCdesign}
\cite{CERNaccelerator}
\cite{LHCschedule}

\cite{RunIIIperformance}
\cite{Polarity}
\cite{LHCb-Performance} 
\cite{LHCb-Run2Performance}
\cite{LHCb-TDR-OG}

\cite{LHCb-LHC}
\cite{LHCb-TDR-TRIG-UPGR}
\cite{LHCb-TDR-SMOG-UPGR}
\cite{LHCb-TDR-018}
\cite{LHCb-PII-Physics}

\cite{LHCb-PII-EoI}
\cite{LHCb-TDR-SOFT}
\cite{LHCb-TDR-ONLINE}
\cite{LHCb-TDR-TRACK-UPGR}
\cite{LHCb-TDR-PID-UPGR}

\cite{LHCb-TDR-VELO-UPGR}
\cite{LHCb-TDR-FRAM}
\cite{LHCb-TDR-COMP}
\cite{LHCb-TDR-TRIG}
\cite{LHCb-TDR-NEW}

\cite{LHCb-TDR-IT}
\cite{LHCb-TDR-DAQ}
\cite{LHCb-TDR-OT}
\cite{LHCb-TDR-TT}
\cite{LHCb-TDR-VELO}

\cite{LHCb-TDR-MUON}
\cite{LHCb-TDR-RICH}
\cite{LHCb-TDR-CALO}
\cite{LHCb-TDR-MAG}
\cite{LHCb-beamMon}
	
\cite{TriggerResources}
\cite{bbangles}    
\cite{LHCb-layout}
\cite{Xi2_states}
\cite{Pentaquark}

\cite{Pentaquark2}
\cite{Tetraquark}
\cite{Wev}
\cite{WandZ}  
\cite{Ztobb} 
	
\cite{SMOG}
\cite{IFT-Antiproton}
\cite{IFT-bottomium}
\cite{LHCb-trigger-diagram}    
\cite{Monte-Carlo}  

\cite{Pythia1}
\cite{Pythia2}
\cite{LHCb-Gauss}
\cite{Geant1}
\cite{Geant2}
		
\cite{EvtGen}
\cite{Brunel}
\cite{Boole}
\cite{Moore}
\cite{CERNmission}        
				
\cite{QQproduction}
\cite{EM-jpsi}
\cite{EM-charm}


\section{Analysis}

\cite{ANA-Buchalla}
\cite{ANA-LHCb_BR}
\cite{ANA-LHCb_P5_AnaNote}
\cite{ANA-Belle_P5}
\cite{ANA-ATLAS_P5}

\cite{ANA-CMS_P5}
\cite{ANA-KstMuMuDecayEvidence}
\cite{ANA-BaBarAngDist}  
\cite{ANA-BelleBF}
\cite{ANA-CDFAngDist}

\cite{ANA-BelleAngDist}
\cite{ANA-LHCb-angular1}
\cite{ANA-LHCb-angular2}
\cite{ANA-LHCb-angular3}
\cite{ANA-LHCb-angular4}

\cite{ANA-BR-Belle}
\cite{ANA-BR-CDF}
\cite{ANA-BR-LHCb}
\cite{ANA-BR-BPHI-LHCb}	
\cite{ANA-BR-BPHI-LHCb2}

\cite{ANA-P5-theo1}
\cite{ANA-P5-theo2}
\cite{ANA-P5-theo3}
\cite{ANA-P5-theo4}	
\cite{ANA-P5-theo5}

\cite{ANA-LHCb-RD}
\cite{ANA-LHCb-Rjpsi}
\cite{ANA-piz-reco}
\cite{ANA-piz-reco2}
\cite{ANA-CALO}

\cite{ANA-CALO-flow}
\cite{ANA-MuonTrig}
\cite{ANA-SingleTrackTrig}
\cite{ANA-HLT2Topo}
\cite{ANA-TrigAll}

\cite{ANA-IsMuon}
\cite{ANA-DTF}
\cite{ANA-MVA}
\cite{ANA-MLP}
\cite{ANA-TMVA}

\cite{ANA-flavio}
\cite{ANA-CMS-angular}
\cite{ANA-Bs-angular}
\cite{ANA-wilsonJi}
\cite{ANA-wilsonNP}

\cite{ANA-Moriond2017}
\cite{ANA-data}


\section{Fit}

\cite{FIT-legendre}
\cite{FIT-bayes}
\cite{FIT-taylor}
\cite{FIT-TMINUIT}
\cite{FIT-BreitWigner}

\cite{FIT-BlattWeisskopf}
\cite{FIT-dSize}
\cite{FIT-JpsiRho}
\cite{FIT-Z}
\cite{FIT-LASS}


\cite{FIT-Dunwoodie}
\cite{FIT-Chebyshev}
\cite{FIT-btosllBall}
\cite{FIT-BaBar}
\cite{FIT-Belle}

\cite{FIT-LHCb-Jpsi}
\cite{FIT-MersenneTwister}
\cite{FIT-testU01}
\cite{FIT-DieHard}
\cite{FIT-NR}

\cite{FIT-hesse}
\cite{FIT-Feldman}


\section{Track}

\cite{Kalman}
\cite{Ghosts}
\cite{tagAndProbe_CMS_1}
\cite{tagAndProbe_CMS_2}
\cite{tagAndProbe_ATLAS}

\cite{TrackEffElectrons}
\cite{Tesla}
\cite{UPC-AnaNote}
\cite{Twiki-TrackCalib}
\cite{TrackEff-FIG}

\section{App}

\cite{APP-CB}
\cite{APP-CBCollab}
\cite{APP-CBMainz}


